Author's response to reviews

Title: A Primary Care, EHR-Based Strategy to Promote Safe and Appropriate Drug Use

Authors:

Bruce L Lambert (bruce.lambert@northwestern.edu)
Kamila Przytula (kamila.przytula@northwestern.edu)
Stacy C Bailey (scbailey@email.unc.edu)
Neeha Shrestha (neeha.shrestha@northwestern.edu)
William G Galanter (billg@uic.edu)
Carolyn Dickens (cdickens@uic.edu)
Suzanne Falck (sfalck@uic.edu)
Michael S Wolf (mswolf@northwestern.edu)

Version: 2
Date: 17 July 2014

Author's response to reviews: see over
July 17, 2014

Dear Editorial Board,

Please consider the revised manuscript entitled, A Primary Care, EHR-Based Strategy to Promote Safe and Appropriate Drug Use, for publication in Trials.

We have made the requested edits to the formatting of the document, as well as provided the additional information regarding authorship, funding, and trial status.

Neither the entire paper nor any of its content has been submitted, published, or accepted by another journal. The paper will not be submitted elsewhere if accepted for publication in Trials.

The corresponding author for this manuscript is:

Bruce L. Lambert, Ph.D.
Professor, Dept. of Communication Studies
Director, Center for Communication and Health
Northwestern University
710 N. Lake Shore Drive, Rm. 414
Chicago, IL 60611
Phone: 312-503-5962
bruce.lambert@northwestern.edu

All authors have reviewed and approved the manuscript, as well as contributed to its content.

We hope that our methodology will be of interest to the readership of Trials and we look forward to hearing from you.

Sincerely,

Kamila Przytula, MS